Himmelfarb Health Sciences Library, The George Washington University Health Sciences Research Commons

Microbiology, Immunology, and Tropical Medicine Faculty Publications Microbiology, Immunology, and Tropical Medicine

5-2012

Elevated plasma IL-6 associates with increased risk of advanced fibrosis and cholangiocarcinoma in individuals infected by Opisthorchis viverrini

Banchob Sripa Khon Kaen University School of Medicine, Khon Kaen, Thailand

Bandit Thinkhamrop Khon Kaen University School of Medicine, Khon Kaen, Thailand

Eimorn Mairiang Khon Kaen University School of Medicine, Khon Kaen, Thailand

Thewarach Laha Khon Kaen University School of Medicine, Khon Kaen, Thailand

Sasithorn Kaewkes Khon Kaen University School of Medicine, Khon Kaen, Thailand

See next page for additional authors

Follow this and additional works at: http://hsrc.himmelfarb.gwu.edu/smhs_microbio_facpubs Part of the <u>Medical Immunology Commons</u>, <u>Medical Microbiology Commons</u>, <u>Parasitic</u> <u>Diseases Commons</u>, and the <u>Parasitology Commons</u>

Recommended Citation

Sripa, B., Thinkhamrop, B., Mairiang, E., Laha, T., Kaewkes, S., Sithithaworn, P., Periago, M.V., Bhudhisawasdi, V., Yonglitthipagon, P., Mulvenna, J., Brindley, P.J., Loukas, A., Bethony, J. M. (2012). Elevated plasma IL-6 associates with increased risk of advanced fibrosis and cholangiocarcinoma in individuals infected by opisthorchis viverrini. PLoS Neglected Tropical Diseases, 6(5):e1654.

This Journal Article is brought to you for free and open access by the Microbiology, Immunology, and Tropical Medicine at Health Sciences Research Commons. It has been accepted for inclusion in Microbiology, Immunology, and Tropical Medicine Faculty Publications by an authorized administrator of Health Sciences Research Commons. For more information, please contact hsrc@gwu.edu.

Authors

Banchob Sripa, Bandit Thinkhamrop, Eimorn Mairiang, Thewarach Laha, Sasithorn Kaewkes, Paiboon Sithithaworn, Maria V. Periago, Vajarabhongsa Bhudhisawasdi, Ponlapat Yonglitthipagon, Jason Mulvenna, Paul J. Brindley, Alex Loukas, and Jeffrey M. Bethony

Elevated Plasma IL-6 Associates with Increased Risk of Advanced Fibrosis and Cholangiocarcinoma in Individuals Infected by *Opisthorchis viverrini*

Banchob Sripa¹, Bandit Thinkhamrop², Eimorn Mairiang³, Thewarach Laha⁴, Sasithorn Kaewkes⁴, Paiboon Sithithaworn⁴, Maria Victoria Periago⁵, Vajarabhongsa Bhudhisawasdi^{6,7}, Ponlapat Yonglitthipagon^{1,8}, Jason Mulvenna⁸, Paul J. Brindley⁹, Alex Loukas⁸, Jeffrey M. Bethony^{5,9}*

1 Department of Pathology, Khon Kaen University School of Medicine, Khon Kaen, Thailand, 2 Department of Epidemiology and Biostatistics, Khon Kaen University School of Public Health, Khon Kaen, Thailand, 3 Department of Radiology, Khon Kaen University School of Medicine, Khon Kaen, Thailand, 4 Department of Parasitology, Khon Kaen University School of Medicine, Khon Kaen, Thailand, 5 Instituto René Rachou, Belo Horizonte, Minas Gerais, Brazil, 6 Department of Surgery, Khon Kaen University School of Medicine, Khon Kaen, Thailand, 7 Liver Fluke and Cholangiocarcinoma Research Center, Khon Kaen University, Khon Kaen, Thailand, 8 Queensland Tropical Health Alliance, James Cook University, Cairns, Queensland, Australia, 9 Department of Microbiology, Immunology & Tropical Medicine, George Washington University Medical Center, Washington, D. C., United States of America

Abstract

Opisthorchis viverrini is considered among the most important of the food-borne trematodes due to its strong association with advanced periductal fibrosis and bile duct cancer (cholangiocarcinoma). We investigated the relationship between plasma levels of Interleukin (IL)-6 and the risk of developing advanced fibrosis and bile duct cancer from chronic Opisthorchis infection. We show that IL-6 circulates in plasma at concentrations 58 times higher in individuals with advanced fibrosis than age, sex, and nearest-neighbor matched controls and 221 times higher in individuals with bile duct cancer than controls. We also observed a dose-response relationship between increasing levels of plasma IL-6 and increasing risk of advanced fibrosis and bile duct cancer; for example, in age and sex adjusted analyses, individuals with the highest quartiles of plasma IL-6 had a 19 times greater risk of developing advanced periductal fibrosis and a 150 times greater risk of developing of bile duct cancer than individuals with no detectable level of plasma IL-6. Finally, we show that a single plasma IL-6 measurement has excellent positive predictive value for the detection of both advanced bile duct fibrosis and bile duct tumorogenesis from chronic *O. viverrini* infection. Our study also adds a unique aspect to the literature on circulating levels of IL-6 as an immune marker of hepatobiliary pathology by showing that high levels of circulating IL-6 in plasma are not related to infection with *O. viverrini*, but to the development of the advanced and often lethal pathologies resulting from chronic *O. viverrini* infection.

Citation: Sripa B, Thinkhamrop B, Mairiang E, Laha T, Kaewkes S, et al. (2012) Elevated Plasma IL-6 Associates with Increased Risk of Advanced Fibrosis and Cholangiocarcinoma in Individuals Infected by *Opisthorchis viverrini*. PLoS Negl Trop Dis 6(5): e1654. doi:10.1371/journal.pntd.0001654

Editor: Edgar M. Carvalho, Hospital Universitário, Brazil

Received July 28, 2011; Accepted March 12, 2012; Published May 22, 2012

Copyright: © 2012 Sripa et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: This research was supported by award number UO1AI065871 from the National Institute of Allergy and Infectious Diseases (NIAID) and R01CA155297-01 from the National Cancer Institute (NCI), both of the National Institutes of Health (NIH), USA. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: jbethony@email.gwu.edu

Introduction

Over 750 million people (10% of the human population) are at risk of infection with a food-borne trematode, with more than 40 million people currently infected [1]. *Opisthorchis viverrini* is considered among the most important of the food-borne trematodes due to its strong association with hepatobiliary pathologies that include advanced bile duct fibrosis (periductal fibrosis or APF) [2–5] and cholangiocarcinoma (CCA) [3,6–11]. In Thailand, an estimated 10 million people are infected with *O. viverrini* [12], where uncooked cyprinoid fish (the intermediate host for the parasite) are a staple of the diet [13]. While infection with *O. viverrini* can be eliminated by chemotherapy (the anthelmintic praziquantel), culinary practices in Thailand result in rapid and prolonged re-infection after treatment. In the Northeastern region of Thailand (Isaan), individuals are often infected with *O. viverrini* for a lifetime [7,14,15].

As shown in our community-based ultrasound studies along the Chi River basin in Isaan (Khon Kaen, Thailand) [16,17], important pathogenic changes occur early and asymptotically to the bile duct during O. viverrini infection, including fibrosis in the intrahepatic bile duct (periductal fibrosis). As with other forms of hepatic fibrosis, periductal fibrosis from chronic opisthorchiasis is probably the result of repeated injury sustained by the biliary epithelium from a combination of the mechanical, toxic, and immune mechanisms of the fluke in the bile duct [for review see [18]]. As individuals are infected with O. viverrini for many years (often a lifetime), a persistent cycle of tissue damage and repair takes place in the intrahepatic biliary ducts, creating a chronic inflammatory milieu that stimulates periductal fibrogenesis [for review of this process see [18,19]]. In both the animal and human models of chronic O. viverrini infection, this fibrotic deposition along the biliary epithelium is a precursor event to CCA. For

Author Summary

O. viverrini is among the few parasites considered a Class 1 carcinogen because of its strong association with bile duct cancer (cholangiocarcinoma). Currently, more than 40 million people are infected with O. viverrini worldwide. Thailand has the highest prevalence of O. viverrini at 10 million people infected and also the highest incidence of Opisthorchis-associated bile duct cancer (cholangiocarcinoma) in the world. In the current study, we also show that levels of IL-6 in plasma are associated in a dose-dependent fashion with Opisthorchis-induced bile duct fibrosis (periductal fibrosis) and cholangiocarcinoma. More importantly, we show that O. viverrini infection alone does not elevate IL-6 levels in the plasma. It is only in the presence of these advanced pathologies (advanced fibrosis or cholangiocarcinoma) from chronic O. viverrini infection that significantly elevates plasma levels of IL-6 are observed. Moreover, we show that plasma IL-6 is an easily accessible biomarker for the detection of advanced periductal fibrosis and cholangiocarcinoma, which would be a critical advance for this region of Thailand and other countries in Southeast Asia, where the prevalence of O. viverrini infection can reach as high as 80% and the incidence of bile duct cancer is the highest in the world.

example, during the final phase of *O. viverini* infection in the hamster model (at 12 weeks after infection), the inflamed biliary epithelium manifests fibrosis throughout its length, which is followed rapidly by tumorogenesis [20,21]. In human autopsy studies, extensive fibrotic deposition in the intrahepatic bile duct routinely accompanies CCA tumors [10,22,23]. Although the exact mechanism has yet to be determined, advanced fibrotic lesions may precede tumorogenesis in the bile duct by producing a "smoldering and chronic inflammatory milieu" [24,25]. A key factor in the maintenance of this chronic inflammatory milieu is the production of soluble growth factors such as the cytokine IL-6 [24,25]. Studies of other hepatic pathogens also strongly implicate

IL-6 in progressive pathogenic fibrosis and carcinogenesis, including hepatocellular fibrosis and hepatocellular carcinoma (HCC) from Hepatitis B (HBV) and Hepatitis C Virus (HCV) infection [26,27].

Due to its role in systemic inflammation, IL-6 is readily detected in plasma [28]. As such, we sought to determine if the concentration of IL-6 in the plasma of *O. viverrini* infected individuals with APF and CCA was higher than age, sex, and nearest-neighbor matched controls infected with *O. viverrini* but without these advanced pathologies. We also sought to determine the sensitivity and specificity of a single measurement of plasma IL-6 to detect APF or CCA in these same individuals. Given the poor prognosis associated with CCA, especially in resource-poor settings such as Thailand, an early marker for the risk of the hepatobiliary pathologies related to *O. viverrini* infection is urgently needed.

Materials and Methods

Study design and setting

The current study presents baseline data collected from a community-based cohort study of the risk factors associated with the development of Advanced Periductal Fibrosis (APF) from chronic opisthorchiasis (Table 1). A detailed description of this population and the methods can be found in [17]. Individuals from seven villages with high O. viverrini transmission, including Nongnangkwan, Nongmuang, Loopka, Nongkham, Nongno, Lawa and Chikokor in the vicinity of Khon Kaen (Thailand) were surveyed. From this group of O. viverrini-infected individuals, 184 males and 236 females between the ages of 20 and 60 years (inclusive) were enrolled into the study as "cases" or age, sex, and nearest neighbor matched "controls". Individuals were enrolled as "cases" on the basis of an ultrasound (US) in which APF was determined [17]. Individuals who were positive for O. viverrini but negative for APF by US were defined as "controls" and matched with cases by age (ten year age intervals), sex, and residence in the same village (nearest-neighbor method). A positive pregnancy test excluded female volunteers from participation in the ultrasonog-

Table 1. Descriptive statistics for cases and controls used in the current study.

		Advanced Periduct			
Characteristics	Non-endemic Controls ¹	Negative Controls ²	Positive Cases ³	CCA Cases ⁴	
	N (%)	N (%)	N (%)	N (%)	
Total	21	210	210	121	
Sex					
Male	11 (52.4%)	92 (43.8%)	92 (43.8%)	83 (68.6%)	
Female	10 (47.6%)	118 (56.2%)	118 (56.2%)	38 (31.4%)	
Age (in years)					
20–29	15 (71.4%)	6 (2.9%)	6 (2.9%)	0 (0.0%)	
30–39	1 (4.8%)	40 (19.0%)	39 (18.6%)	6 (7.2%)	
40–49	3 (14.3%)	86 (41.0%)	87 (41.4%)	20 (24.1%)	
50+	2 (9.5%)	78 (37.1%)	78 (37.1%)	57 (68.7%)	

¹Non-endemic control" refers to age-matched Thai individuals who have never resided in an area with *O. viverrini* transmission.

²A control refers to an *O. viverrini*-infected individual who is age, sex, and nearest neighbor matched to a "case" but are negative for Advanced Periductal Fibrosis. ³A case refers to *O. viverrini* infected individual who is positive for Advanced Periductal Fibrosis.

⁴A case in this column refers to an individual with histologically proven *O. viverrini* associated cholangiocarcinoma (CCA) from the biological repository of the Liver Fluke and Cholangiocarcinoma Research Center, Faculty of Medicine, Khon Kaen University, Thailand.

doi:10.1371/journal.pntd.0001654.t001

raphy and blood draw. As such, 210 individuals were identified as cases and matched 210 individuals as controls. Both cases and control were asked to provide 30 ml of blood for baseline immunology and hematological parameters. Individuals positive for *O. viverini* were referred to the local public health outpost for treatment with praziquantel.

All subjects provided written informed consent using Informed Consents Forms approved by the Ethics Committee of Khon Kaen University School of Medicine, Khon Kaen, Thailand (reference number HE480528) and the Institutional Review Board of the George Washington University School of Medicine, Washington, D.C (GWUMC IRB# 020864).

CCA cases

Aliquots of plasma from 121 cases of histologically proven, *O. viverrini* associated CCA cases were pulled from the biological repository of the Liver Fluke and Cholangiocarcinoma Research Center, Faculty of Medicine, Khon Kaen University, Thailand (Table 1). Of these pulled samples, 83 were male and 38 were female. All samples were from patients who had liver resection surgery as a part of palliative care for *O. viverrini*-associated CCA at the Khon Kaen University Srinagarind Hospital, Khon Kaen, Thailand.

Ultrasonography

A detailed description of the ultrasonography methods used in this study can be found in the following references [16,17]. Briefly, a mobile, high-resolution ultrasound (US) machine (GE model LOGIQ Book XP) was used. Hepatobiliary abnormalities including portal vein radical echoes, echoes in liver parenchyma, indistinct gallbladder wall, gallbladder size, sludge and suspected CCA were graded and recorded as previously described [16,17]. Individuals were classified as "Non-Advanced Periductal Fibrosis" or "controls" if the US grade was 0 or 1, and "Advanced Periductal Fibrosis" or "case" if the US grade was 2 or 3 as described in detail in our previous study [16,17]. Individuals with alcoholic liver disease, which is seen as fatty liver by US exam, were excluded from the analysis component of the study. Also, individuals with marked hepatic fibrosis not related to O. viverrini infection (e.g., cirrhosis from HBV or HCV) were also excluded from the analysis component study.

Fecal exams to determine O. viverrini infection status

Eggs/parasite identification and egg counts were performed by certified medical technicians using light microscopy at $10 \times$ and $40 \times$ magnifications as described in detail in our previous study [16].

Plasma handling

Of the approximately 30 ml of blood collected from participants, 8 ml of blood were collected in heparinized tubes for the measurement of plasma cytokines. Plasma aliquots were grouped in case-control sets that were handled together throughout the processing and biochemical analysis. After blood draw, plasma was separated and immediately aliquoted into 500 microliter (mL) cryogenic tubes and frozen at -80° C until use. Biochemical analysis was done on all samples simultaneously, with the position of plasma samples varying at random during the processing. Plasma samples from CCA patients were taken at the time of liver resection, separated and aliquoted as above, and then stored at -80° C until use. CCA plasma samples were also randomly interspersed among the cases and control sample sets during biochemical analysis. The 21 non-endemic *O. viverrini* negative control plasma were handled as above and interspersed repeatedly at random among the actual case-control samples during biochemical analysis.

Exploratory plasma cytokine measurements

Cytokine levels in plasma were examined for IL-1 β , IL-2, IL-4, IL-5, IL-6, IL-8, IL-10, IL-12p70, interferon (INF) γ , tumor necrosis factor (TNF α), and TNF β production. The quantification of plasma cytokines was analyzed using commercial FlowCytomix bead-based multiplexing assays kits (Beckman-Coulter). Values were quantified from standard curves using human recombinant cytokines. A quantile regression model was used to determine the median along with 95% confidence intervals (95% CI) to estimate the differences between cases and controls for plasma cytokine levels. When the data suggested a significant difference between cases and controls, such as the levels of plasma IL-6, the plasma were analyzed again using a sandwich ELISA as described below.

IL-6 ELISA

IL-6 cytokine levels in the plasma were measured by sandwich enzyme-linked immunoadsorbent assay (ELISA) using a DuoSet (R & D systems, Inc) according to the manufacturer's instruction. The level of IL-6 was determined by interpolating the Optical Density of sample duplicates into a 4-parameter logistic-log model of a standard curve of recombinant human IL-6 run in serial dilutions on each ELISA plate. The concentration of IL-6 level is expressed as picograms per microliter.

Statistics for Odd Ratios by quartile of IL-6 plasma concentration

The percent distribution of selected demographic characteristics were calculated for the four study groups: (1) non-endemic negative controls, (2) *O. viverini* positive (OV+) and APF negative (controls), (3) OV+ APF positive (APF cases), and (4) OV+ and CCA+ (CCA cases). Box and whisker plots display the distribution of plasma IL-6 according to these groups. Age and sex adjusted Odds Ratios (OR) and 95% Confidence Intervals (CIs) for quartiles of plasma IL-6 concentration and their association with APF or CCA status were determined using age and sex adjusted multiple logistic regression analyses. A chi-square test for trend was also used to test the effect of increasing quartile level of plasma IL-6 on increasing risk of APF or CCA. The significance level for all tests was set at 0.05, with a Bonferroni correction for multiple testing. All statistical tests were two-sided. All analyses were performed using Stata version 10 (College Station, TX).

Statistics for clinical epidemiology of IL-6 plasma concentration and APF and CCA

Receiver-operating-characteristic (ROC) curves were obtained by plotting the sensitivity versus 1–specificity for the full range of IL-6 cut-off points in picograms per milliliter (pg/mL) for both cases and controls to estimate the cutoff value that had the highest overall validity. "Sensitivity" was calculated as the number of individuals positive for APF or CCA testing positive for plasma IL-6 at various cutpoints (by pg/mL) divided by the total number of cases. "Specificity" was calculated as the number of individuals negative for APF and CCA (controls) testing negative for plasma IL-6 at various cutpoints (by pg/mL) divided by the total number of controls. The area under the ROC curve was calculated by determining the probability of correctly identifying (accuracy) a randomly selected participant as either a case (APF positive or CCA positive) or a non-case (APF negative or CCA negative). The 45-degree line in each ROC curve graph subsumes an area equal to 0.5 (50%), which is equivalent to using a coin toss procedure to classify participants as either cases or controls. Using the ROC curves, an optimal cutpoint was determined for the concentration of plasma IL-6 that maximized the sensitivity and specificity in classifying an individual at APF+ or CCA+. Based on this cutoffpoint, the positive predictive value (PPV), and negative predictive value (NPV) of plasma IL-6 concentration to detected APF+ or CCA+ status were determined. All analyses were performed using Stata version 10 (College Station, TX).

Results

Study participants

As part of the inclusion criteria of the study, both cases and controls had to be positive for O. viverini as determined by the presence of at least a single parasite ovum in feces. The baseline characteristics of the enrolled participants are shown in Table 1, with 210 cases (O. viverrini+ and APF+) and 210 controls (O. viverrini+ and APF-). The controls were matched with cases by age (ten year age intervals), sex, infection (O. viverrini positive) and nearest-neighbor status (same village). The study sample included more females (56.2%) than males (43.8%) (P < 0.001). The mean ages for cases (46.6 years of age) and controls (46.6 years of age) did not differ significantly (P = 0.999). No statistically significant difference (P>0.811) was observed in the intensity of O. viverini infection between cases (median = 142 epg) and controls (median = 130 epg). As shown in Table 1, both case and control groups had a similar distribution of individuals when age was stratified by ten-vear intervals.

Among the 121 CCA samples in this study (Table 1), 117 (96.7%) were hepatectomies and 4 (3.3%) were small biopsy specimens. Of the 117 hepatectomies, 74 (63.2%) were of the

mass-forming type and 8 (6.8%), 21 (18%), and 14 (12%) were periductal-infiltrating, invasive intraductal, and mixed types, respectively. Histologically, there were 41 well-differentiated (33.9%), 8 moderately differentiated (6.6%), 9 poorly differentiated adenocarcinomas (7.4%), 61 papillary carcinomas (50.4%), and 2 adenosquamous carcinomas (1.6%) of the 121 CCA cases studied.

The concentration of plasma IL-6 is significantly elevated in individuals with Opisthorchis-induced APF and Opisthorchis-induced CCA

Of the 11 cytokines tested only levels of the inflammatory cvtokine IL-6 were significantly elevated in individuals with advanced periductal fibrosis and CCA compared to controls. The frequency distributions of plasma IL-6 concentration for all groups are presented in Figure 1. The IL-6 concentrations in plasma for both cases and controls ranged between <0.01 pg/mL (undetectable) and 538.6 pg/ml, with a much wider range in APF cases (<0.01 to 538.6 pg/ml) and CCA cases (<0.01 to 536.2 pg/ ml) than APF- controls (<0.01 to 173.1 pg/ml). An even narrower range of plasma IL-6 concentration was recorded for the nonendemic controls (<0.01 to 91.4 pg/ml). These data indicate that there is far wider variation for the concentration of plasma IL-6 in the advanced forms of O. viverrini induced pathogenesis than in individuals with O. viverrini infection but no O. viverrini associated pathology. Moreover, the median levels of plasma IL-6 concentration were 58 times higher in APF+ cases than in controls (58 versus 1 pg/ml; P<0.001) and 221 times higher in CCA cases than in controls (221 versus 1 pg/ml; P<0.001). CCA cases also had higher levels of plasma IL-6 than APF cases (221 versus 58 pg/ml; P<0.001). Non-endemic controls had significantly lower (mostly undetectable or <0.01 pg/mL) plasma IL-6 concentrations (P<0.0001).



Figure 1. Plasma IL-6 levels are significantly elevated in individuals with Opisthorchis-induced APF and Opisthorchis-induced CCA. The distribution of plasma concentration of Interleukin (IL)-6) in picograms per milliliter is shown in boxplots by study group. The bottom and top of each box represent the 25th and 75th percentile (the lower and upper quartiles, respectively) of IL-6 concentration in plasma per group. The band near the middle of each box represents the median value of IL-6 concentration per group. The whiskers represent the minimum (bottom) and maximum (top) of IL-6 concentration per group. doi:10.1371/journal.pntd.0001654.g001

Table 2. Odds Ratios for plasma IL-6 levels for O. viverrini infected individuals with and without Advanced Periductal Fibrosis.

	Quartile IL-6					
	11	2	3	4	P for Trend	
	(<0.01)	(0.01 to 21.62)	(21.63 to 82.79)	(82.80 to 358.63)		
n	180	76	128	127		
Crude					P=<0.001	
OR ²	1.00	3.88	8.02	18.55		
95% Cl ³	-	2.21-6.81	4.78-13.46	9.99–34.42		
Р	-	<0.001	<0.001	<0.001		
Adjusted ⁴					NA ⁵	
OR	1.00	3.94	7.95	18.94		
95% CI	-	2.24-6.95	4.73-13.36	10.17-35.25		
Р	-	<0.001	<0.001	<0.001		

¹Individuals with undetectable levels of plasma IL-6 concentration are used as the reference group in Quartile 1.

²Odds Ratio.

³95% Confidence Interval.

⁴Models were adjusted for age and sex simultaneously.

⁵Not Available.

doi:10.1371/journal.pntd.0001654.t002

Elevated levels of plasma IL-6 significantly increase the risk of APF in a dose-dependent manner

Increasing plasma concentrations of IL-6 significantly increased the risk of APF: for every 10 pg/ml increase of IL-6, there is an increase in the risk of APF by 22% (OR = 1.22; 95% CI 1.16 to 1.29, P<0.001) in a model adjusted for age and sex (data not shown). As shown in Table 2, the risk of APF increased with increasing quartile concentration of plasma IL-6: OR = 1.00 for Quartile 1 (reference quartile); OR = 3.94 (95% CI 2.21 to 6.81) for Quartile 2; OR = 7.95 (95% CI 4.73 to 13.36, P = <0.001) for Quartile 3; and OR = 18.94 (95% CI 10.17 to 35.25, P = <0.001) for Quartile 4. A strong and significant trend for increasing quartile concentration of IL-6 and increasing risk of APF was found compared to individuals with no detectable levels of plasma IL-6 (P<0.001).

Elevated levels of plasma IL-6 significantly increase the risk of developing CCA in a dose-dependent manner

Elevated plasma concentrations of IL-6 significantly increased the risk of CCA: for every 10 pg/ml increase of plasma IL-6 concentration, there was a 26% (OR = 1.26; 95% CI 1.19 to 1.34, P<0.001) increase in the risk of CCA in a model adjusted for age and sex. Table 3 shows that the risk of CCA increased in the 3rd and 4th highest quartile concentrations of plasma IL-6: OR = 4.55 (95% CI 2.05 to 10.11, P = <0.001) for Quartile 3 and OR = 149.11 (95% CI 40.42 to 550.15, P = <0.001) for Quartile 4. A strong and significant

Table 3. Odds Ratios for plasma IL-6 levels for O. viverrini infected individuals with¹ and without Cholangiocarcinoma (CCA)².

	Quartile IL-6 (range in pg/ml)					
	1 ³	2	3	4	P for Trend	
	(<0.01)	(0.01 to 21.62)	(21.63 to 82.79)	(82.80 to 358.64)		
Ν	145	21	83	82		
Crude						
OR ⁴	1.00	0.47	4.76	182.46	P = < 0.001	
95% Cl⁵	-	0.06-3.75	2/33-9.73	58.01-573.96		
Р	-	0.475	<0.001	<0.001		
Adjusted ⁶						
OR	1.00	0.21	4.55	149.11	NA ⁷	
95% CI	_	0.02-2.30	2.05-10.11	40.42-550.15		

¹Individual without CCA are the controls from the case-control study in Table 2, that is, *O. viverrini* infected individuals who are negative for Advanced Periductal Fibrosis (APF) by ultrasound and were age, sex, and nearest-neighbor matched with APF positive individuals (see Table 1).

²CCA cases were from the biological repository of the Liver Fluke and Cholangiocarcinoma Research Center, Faculty of Medicine, Khon Kaen University, Thailand. ³Individuals with undetectable levels of plasma IL-6 concentration are used as the reference group in Quartile 1.

⁴Odds Ratio.

⁵95% Confidence Interval.

⁶Models were adjusted for age and sex simultaneously.

⁷Not Available.

doi:10.1371/journal.pntd.0001654.t003

trend (P<0.001) for increasing quartile concentration of IL-6 and increasing risk of CCA was also found compared to individuals with no detectable level of plasma IL-6 (<0.01 pg/mL). Note that the term "control" refers to the age, sex, and nearest-neighbor matched controls for APF group and not for the CCA group and that the logistic model is adjusted for age and sex.

Plasma IL-6 concentrations can be used to detect individuals with Opisthorchis-induced advanced periductal fibrosis (APF)

Figure 2 shows the ROC curve obtained by plotting the True Positive Probability (sensitivity) against the False Negative Probability (1–specificity) for the entire range of IL-6 cut-off points to predict the presence of APF. Using a cutoff of greater than 11 pg/mL of plasma IL-6, the sensitivity for the detection of APF was 80% and the specificity was 74% (Table 4). In addition, for the cutoff of greater than 11 pg/mL of plasma IL-6, the area under the ROC curve, which is an indication of the "accuracy" of the test or proportion of all tests that have given the correct result, was 78% (95%CI 74% to 83%). Table 4 also shows the Positive Predictive Value (PPV) and Negative Predictive Value (NPV) of 76% and 79%, respectively, for the detection of APF using plasma IL-6 levels greater than 11 pg/mL.

Plasma IL-6 concentrations can be used to detect *O. viverrini* infected individuals with cholangiocarcinoma (CCA)

Figure 3 shows the ROC curve obtained by plotting the True Positive Probability (sensitivity) against the False Negative



Probability (1–specificity) for the entire range of IL-6 cut-off points to predict the presence of CCA. Using a cutoff of greater than 64 pg/mL of plasma IL-6, the sensitivity for the detection of CCA was 80% and the specificity was 90% (Table 4). In addition, for the optimal cutoff of >64 pg/mL of plasma IL-6, the area under the ROC curve, which is an indication of the "accuracy" of the test, was 89% (95% CI = 85% to 93%). Table 4 also shows the Positive Predictive Value (PPV) and Negative Predictive Value (NPV) of 82% and 88%, respectively, for the detection of APF using plasma IL-6 levels >64 pg/mL. Note that the term control refers to the age, sex, and nearest neighbor-matched controls for APF group and not for the CCA group.

Discussion

These data show that elevated plasma concentrations of IL-6 are associated with a marked and significant increase in the risk of O. viverrini-associated APF and CCA. As shown in Tables 2 and 3, increasing levels of plasma IL-6 associate with increasing risk of these advanced pathologies in a dose-dependent manner: for example, individuals with the highest quartiles of plasma IL-6 concentration had a 19 times greater risk of developing APF and a 150 times greater risk of developing CCA than individuals with undetectable levels of plasma IL-6 (<0.01 g/mL). As the data were collected by a cross-sectional study design, the complicity of IL-6 in these pathogenic processes remains to be determined: that is, elevated plasma concentrations of IL-6 may simply reflect the presence of these hepatobiliary abnormalities or, as shown in a number of other studies, elevated plasma concentrations of IL-6

Figure 2. Plasma IL-6 concentrations can be used to detect individuals with Opisthorchis-induced advanced periductal fibrosis. A receiver-operating-characteristic (ROC) curve plots the True Positive Probability (sensitivity) against the False Negative Probability (1– specificity) for the full range of IL-6 cut-off points for the detection of Opisthorchis-induced Advanced Periductal Fibrosis (APF) as determined by ultrasound. The area under the ROC curve is interpreted as the probability of correctly identifying (accuracy) a randomly selected participant as either a case (APF positive) or a non-case (APF negative). The 45-degree line in the graph subsumes an area equal to 0.50 (50%), which is equivalent to using a coin toss procedure to classify participants as either cases or controls. As determined by this ROC curve, the optimal cutpoint is 11 pg/mL of plasma IL-6, which refers to the concentration that maximizes the "sensitivity" and the "specificity" of classifying an individual at APF positive. Based on this cutoff point, the Positive Predictive Value (PPV) and the Negative Predictive Value (NPV) of plasma IL-6 concentration to detect APF was also determined (see Table 4). All analyses were performed using Stata version 10 (College Station, TX). The ROC was derived from 210 *O. viverrini* infected individuals with Advanced Periductal Fibrosis as determined by ultrasound (gold standard) versus 210 age, sex, and nearest neighbor-matched controls (*O. viverrini* infected but negative preductive Controls (*D. viverrini* infected but negative preductive Controls (*D. viverrini* infected but negative preducted by ultrasound (gold standard) versus 210 age, sex, and nearest neighbor-matched controls (*O. viverrini* infected but negative preductive cutore of the control of the control section of the control of the preductive cutore of the control section of the contrent of the control section of the contrent of the control

doi:10.1371/journal.pntd.0001654.g002

Table 4. The diagnostic utility of plasma Interleukin (IL)-6 levels for Opisthorchis-induced pathologies.

	Ν			Cut off	Percent (%)					
	Case	Con ¹	Tot	>pg/mL	Sen ²	Spec ³	Acc ⁴	PPV⁵	NPV ⁶	
APF ⁷	210	210	420	11	80	74	77	76	79	
CCA ⁸	121	210	331	64	80	90	86	82	88	

¹Age, sex, and nearest-neighbor matched individuals, who were positive for *O. viverrini* infection and negative for APF as determined by ultrasound (US) were included in the analyses and shown as controls in Table 1.

²Sensitivity.

³Specificity.

⁴Accuracy of the test or the area under the Receiver-Operating Characteristic (ROC) curves in Figures 2 and 3.

⁵Positive Predictive Value.

⁶Negative Predictive Value.

⁷O. viverrini infected individuals with Advanced Periductal Fibrosis as determined by US and shown as cases in Table 1.

⁸Plasma samples from histologically confirmed Opisthorchis-induced cholangiocarcinoma from the Liver Fluke and Cholangiocarcinoma Research Center, Faculty of Medicine, Khon Kaen University, Thailand.

doi:10.1371/journal.pntd.0001654.t004

may play a key role in these pathogenic processes by creating an inflammatory milieu that favors fibrotic deposition and carcinogenesis in the bile duct [for reviews see [28–30]]. Notably, our data show that *O. viverrini* infection alone does not elevate IL-6 levels circulating in the plasma: e.g., age and sex matched *O. viverrini* infected individuals without APF or CCA had negligible levels of IL-6 in their plasma. It is only in the presence of advanced pathology from chronic opisthorchiasis that plasma concentrations of IL-6 are significantly elevated. This study also shows that plasma IL-6 concentration can be used to detect the risk for the advanced pathologies associated with *O. viverrini* infection, many of which are subclinical. Our data on the predictive values of a single measurement of plasma IL-6 to detect CCA closely resemble recent data from Cheon and coworkers [31] as well as other studies that have also reported a high sensitivity and specificity for IL-6 in serum for non-Opisthorchisassociated CCA [see [32] for review]. Our study adds a unique aspect to the literature on circulating levels of IL-6 as an immune



Figure 3. Plasma IL-6 concentrations can be used to detect O. viverrini infected individuals with cholangiocarcinoma (CCA). A receiver-operating-characteristic (ROC) curve plots the True Positive Probability (sensitivity) against the False Negative Probability (1– specificity) for the full range of IL-6 cut-off points for the determination of Opisthorchis-induced cholangiocarcinoma (CCA). The area under the ROC curve is interpreted as the probability of correctly identifying (accuracy) a randomly selected participant as either a case (APF positive) or a non-case (APF negative). The 45-degree line in the graph subsumes an area equal to 0.50 (50%), which is equivalent to using a coin toss procedure to classify participants as either cases or controls. As determined by this ROC curve, the optimal cutpoint is 11 pg/mL of plasma IL-6, which refers to the concentration that maximizes the "sensitivity" and the "specificity" of classifying an individual at APF positive. Based on this cutoff point, the Positive Predictive Value (PPV) and the Negative Predictive Value (NPV) of plasma IL-6 concentration to detect APF was also determined (see Table 4). All analyses were performed using Stata version 10 (College Station, TX). The ROC was derived from 121 cases of histologically proven, *O. viverrini* associated CCA cases from the biological repository of the Liver Fluke and Cholangiocarcinoma Research Center, Faculty of Medicine, Khon Kaen University, Thailand and 210 individuals who are *O. viverrini* infected but negative for APF and CCA.

. www.plosntds.org

marker of hepatobiliary pathology by showing that high levels of circulating IL-6 in plasma are not related to infection with O. viverrini, but to the development of the advanced and often lethal pathologies resulting from chronic O. viverrini infection: i.e., age and sex matched controls with O. viverrini infection and no pathology (APF- and CCA-) had undetectable levels of this inflammatory cytokine in their plasma. Moreover, the ability of a single IL-6 measurement to detect risk for O. viverrini associated CCA and O. viverrini associated pathogenic fibrosis in the bile duct (APF) is especially important in regions where O. viverrini is endemic. In Isaan, Thailand, for example, the prevalence of O. viverrini infection can reach as high as 79%; hence, an easily accessible immune marker which can distinguish between infection with O. viverini and the advanced pathology induced by this parasite would be particularly useful, where the incidence of intraheptic CCA is among the highest in the world [33]. Currently, Thai individuals in O. viverrini endemic areas are diagnosed only in the most advanced stages of CCA, when treatment is essentially palliative. An easily measurable biomarker for a precursor stage to CCA such as APF would be of great public health importance by identifying those at greatest risk of CCA.

IL-6 is produced by numerous cell types, with a broad range of effects, including the promotion of the innate immune response to pathogens as well as the subsequent chronic inflammatory reactions [28,30,34]. Once known as "hepatocyte-stimulating factor" (HSF), IL-6 has been shown to play a key role in chronic inflammatory conditions of the liver that lead to fibrotic lesions, e.g., chronic alcohol consumption or viral hepatitis [26,35]. Our previous studies have shown that crude antigens extracts antigens from this parasite stimulate host PBMC from O. viverrini infected individuals to produce high levels of IL-6 [16]. Our new data indicate that elevated levels of IL-6 are not confined to the immune response to parasite antigens, but circulate at elevated levels in the plasma of individuals with advanced pathologies associated with chronic O. viverrini infection, a role consistent with the reputation of IL-6 as a key player in systemic inflammation. We speculate that elevated concentrations of IL-6 circulating in the plasma reflect the constant tissue repair response of the biliary epithelia to the chronic mechanical, toxic, and immune injury induced by the fluke [18]. When these tissue repair mechanisms are activated transiently, the normal hepatobiliary structure and function rapidly recovers: i.e., cholangiocytes would regenerate and replace the necrotic or apoptotic cells, a process associated with a local and transitory IL-6 response. However, as with chronic alcohol consumption or infection with either HBC or HCV infection, chronic opisthorchiasis creates a persistent

References

- Keiser J, Utzinger J (2005) Emerging foodborne trematodiasis. Emerg Infect Dis 11: 1507–1514.
- Mairiang E, Elkins DB, Mairiang P, Chaiyakum J, Chamadol N, et al. (1992) Relationship between intensity of *Opisthorchis viverrini* infection and hepatobiliary disease detected by ultrasonography. J Gastroenterol Hepatol 7: 17–21.
- Elkins DB, Mairiang E, Sithithaworn P, Mairiang P, Chaiyakum J, et al. (1996) Cross-sectional patterns of hepatobiliary abnormalities and possible precursor conditions of cholangiocarcinoma associated with *Opisthorchis viverrini* infection in humans. Am J Trop Med Hyg 55: 295–301.
- Mairiang E, Mairiang P (2003) Clinical manifestation of opisthorchiasis and treatment. Acta Trop 88: 221–227.
- Mairiang E, Chaiyakum J, Chamadol N, Laopaiboon V, Srinakarin J, et al. (2006) Ultrasound screening for *Opisthorchis viverrini*-associated cholangiocarcinomas: experience in an endemic area. Asian Pac J Cancer Prev 7: 431–433.
- Elkins DB, Haswell-Elkins MR, Mairiang E, Mairiang P, Sithithaworn P, et al. (1990) A high frequency of hepatobiliary disease and suspected cholangiocarcinoma associated with heavy *Opisthorchis viverrini* infection in a small community in north-east Thailand. Trans R Soc Trop Med Hyg 84: 715–719.
- Haswell-Elkins MR, Sithithaworn P, Elkins D (1992) Opisthorchis viverrini and cholangiocarcinoma in Northeast Thailand. Parasitol Today 8: 86–89.

inflammatory milieu that stimulates fibrotic deposition [25,29,36], a process that could be reflected or even partially induced by the elevated concentrations of IL-6 circulating in the plasma.

As with other inflammation associated hepatic pathologies [25,29,36,37], tumors in the biliary tract may have activated the wound-healing programs of the host chronically infected with O. viverrini in an exaggerated and prolonged manner. A parallel scenario is the presence of pathogenic fibrosis before the development of HCC in patients with chronic HCV [for review see [27]]. In the case of HCC, it is hypothesized that the development of fibrosis and/or cirrhosis, plus a microenvironment conducive to genomic instability, promotes neoplastic transformation in areas of hepatic fibrosis. Prominent among these processes is the production of soluble growth factors such as IL-6 [25,29,36]. In the hamster animal model of O. viverrini-induced CCA, the constant injury to bile duct epithelium by the liver fluke engages a similar process [11]. During the early stage of infection (4 weeks), fluke injury to the biliary epithelium results in inflammatory cell infiltration, hyperplasia, and in the adenomatous change in the bile duct epithelium. By the chronic stage of the infection (12 weeks), the immune response has transformed into prolonged inflammation with fibrotic deposition along the bile duct wall and ultimately CCA [18,38-41]. In autopsy studies [22,23] and studies on liver tissue resected from Opisthorchis-induced CCA patients for palliative care [5,18], fibrosis is routinely detected proximal to neoplastic bile duct tissue.

In summary, this study identifies a significant relationship between plasma IL-6 concentration and the advanced pathologies associated with chronic *O. viverrini* infection. While in other settings (e.g. Western countries) the causative agent of CCA remains obscure, the single most important risk factor for intrahepatic CCA in Thailand has long been established–infection with the liver fluke *O. viverrini* [9]. As such, an easily accessible biomarker such as plasma IL-6 would have great utility in predicting those at risk (APF) or already with early CCA in a setting, where half of the population are routinely exposed to this class I carcinogen *O. viverrini* through the daily consumption of raw fish [42].

Author Contributions

Conceived and designed the experiments: BS BT EM PS PJB AL JMB. Performed the experiments: BS EM JMB. Analyzed the data: BS BT EM PJB AL JM JMB. Contributed reagents/materials/analysis tools: TL SK MVP VB PY. Wrote the paper: BS BT AL JMB.

- Haswell-Elkins MR, Mairiang E, Mairiang P, Chaiyakum J, Chamadol N, et al. (1994) Cross-sectional study of *Opisthorchis viverrini* infection and cholangiocarcinoma in communities within a high-risk area in northeast Thailand. Int J Cancer 59: 505–509.
- IARC (1994) Schistosomes, Liver Flukes and Helicobacter Pylori. Internatinal Agency for Research on Cancer World Health Organization.
- Sithithaworn P, Haswell-Elkins MR, Mairiang P, Satarug S, Mairiang E, et al. (1994) Parasite-associated morbidity: liver fluke infection and bile duct cancer in northeast Thailand. Int J Parasitol 24: 833–843.
- Sripa B, Kaewkes S, Sithithaworn P, Mairiang E, Laha T, et al. (2007) Liver fluke induces cholangiocarcinoma. PLoS Med 4: e201.
- Sithithaworn P, Haswell-Elkins M (2003) Epidemiology of Opisthorchis viverrini. Acta Trop 88: 187–194.
- Sithithaworn P, Pipitgool V, Srisawangwong T, Elkins DB, Haswell-Elkins MR (1997) Seasonal variation of *Opisthorchis viverrini* infection in cyprinoid fish in north-east Thailand: implications for parasite control and food safety. Bull World Health Organ 75: 125–131.
- Haswell-Elkins MR, Elkins DB, Sithithaworn P, Treesarawat P, Kaewkes S (1991) Distribution patterns of Opisthorchis viverrini within a human community. Parasitology 103 Pt 1: 97–101.

- Mairiang E, Haswell-Elkins MR, Mairiang P, Sithithaworn P, Elkins DB (1993) Reversal of biliary tract abnormalities associated with Opisthorchis viverrini infection following praziquantel treatment. Trans R Soc Trop Med Hyg 87: 194–197.
- Sripa B, Mairiang E, Thinkhamrop B, Laha T, Kaewkes S, et al. (2009) Advanced periductal fibrosis from infection with the carcinogenic human liver fluke *Opisthorchis viverrini* correlates with elevated levels of interleukin-6. Hepatology 50: 1273–1281.
- Mairiang E, Laha T, Bethony JM, Thinkhamrop B, Kaewkes S, et al. (2011) Ultrasonography assessment of hepatobiliary abnormalities in 3359 subjects with *Opisthorchis viverrini* infection in endemic areas of Thailand. Parasitol Int.
- Sripa B (2003) Pathobiology of opisthorchiasis: an update. Acta Trop 88: 209–220.
- 19. Bataller R, Brenner DA (2005) Liver fibrosis. J Clin Invest 115: 209-218.
- Thamavit W, Bhamarapravati N, Sahaphong S, Vajrasthira S, Angsubhakorn S (1978) Effects of dimethylnitrosamine on induction of cholangiocarcinoma in *Opisthorchis viverrini*-infected Syrian golden hamsters. Cancer Res 38: 4634–4639.
- Thamavit W, Moore MA, Hiasa Y, Ito N (1988) Enhancement of DHPN induced hepatocellular, cholangiocellular and pancreatic carcinogenesis by *Opisthorchis viverrini* infestation in Syrian golden hamsters. Carcinogenesis 9: 1095–1098.
- Sithithaworn P, Tesana S, Pipitgool V, Kaewkes S, Pairojkul C, et al. (1991) Relationship between faecal egg count and worm burden of *Opisthorchis viverrini* in human autopsy cases. Parasitology 102 Pt 2: 277–281.
- Sithithaworn P, Tesana S, Pipitgool V, Kaewkes S, Thaiklar K, et al. (1991) Quantitative post-mortem study of *Opisthorchis viverrini* in man in north-east Thailand. Trans R Soc Trop Med Hyg 85: 765–768.
- 24. Dvorak HF (1986) Tumors: wounds that do not heal. Similarities between tumor stroma generation and wound healing. N Engl J Med 315: 1650–1659.
- Schafer M, Werner S (2008) Cancer as an overhealing wound: an old hypothesis revisited. Nat Rev Mol Cell Biol 9: 628–638.
- Sheikh MY, Choi J, Qadri I, Friedman JE, Sanyal AJ (2008) Hepatitis C virus infection: molecular pathways to metabolic syndrome. Hepatology 47: 2127–2133.
- Castello G, Scala S, Palmieri G, Curley SA, Izzo F (2010) HCV-related hepatocellular carcinoma: From chronic inflammation to cancer. Clin Immunol 134: 237–250.
- Nishimoto N, Kishimoto T (2006) Interleukin 6: from bench to bedside. Nat Clin Pract Rheumatol 2: 619–626.

Plasma IL-6 and O. viverrini-Induced Pathogenesis

- Balkwill F, Charles KA, Mantovani A (2005) Smoldering and polarized inflammation in the initiation and promotion of malignant disease. Cancer Cell 7: 211–217.
- Heikkila K, Ebrahim S, Lawlor DA (2008) Systematic review of the association between circulating interleukin-6 (IL-6) and cancer. Eur J Cancer 44: 937–945.
- Cheon YK, Cho YD, Moon JH, Jang JY, Kim YS, et al. (2007) Diagnostic utility of interleukin-6 (IL-6) for primary bile duct cancer and changes in serum IL-6 levels following photodynamic therapy. Am J Gastroenterol 102: 2164–2170.
- Goydos JS, Brumfield AM, Frezza E, Booth A, Lotze MT, et al. (1998) Marked elevation of serum interleukin-6 in patients with cholangiocarcinoma: validation of utility as a clinical marker. Ann Surg 227: 398–404.
- Parkin DM (2006) The global health burden of infection-associated cancers in the year 2002. Int J Cancer 118: 3030–3044.
- Naugler WE, Karin M (2008) The wolf in sheep's clothing: the role of interleukin-6 in immunity, inflammation and cancer. Trends Mol Med 14: 109–119.
- McClain CJ, Song Z, Barve SS, Hill DB, Deaciuc I (2004) Recent advances in alcoholic liver disease. IV. Dysregulated cytokine metabolism in alcoholic liver disease. Am J Physiol Gastrointest Liver Physiol 287: G497–502.
- de Visser KE, Eichten A, Coussens LM (2006) Paradoxical roles of the immune system during cancer development. Nat Rev Cancer 6: 24–37.
- Shimoda M, Mellody KT, Orimo A (2010) Carcinoma-associated fibroblasts are a rate-limiting determinant for tumour progression. Semin Cell Dev Biol 21: 19–25.
- Sripa B, Kaewkes S (2000) Localisation of parasite antigens and inflammatory responses in experimental opisthorchiasis. Int J Parasitol 30: 735–740.
- Sripa B, Kaewkes S (2000) Relationship between parasite-specific antibody responses and intensity of *Opisthorchis viverrini* infection in hamsters. Parasite Immunol 22: 139–145.
- Sripa B, Kaewkes S (2002) Gall bladder and extrahepatic bile duct changes in Opisthorchis viverrini-infected hamsters. Acta Trop 83: 29–36.
- Pinlaor S, Sripa B, Sithithaworn P, Yongvanit P (2004) Hepatobiliary changes, antibody response, and alteration of liver enzymes in hamsters re-infected with *Opisthorchis viverrini*. Exp Parasitol 108: 32–39.
- 42. Sripa B, Bethony JM, Sithithaworn P, Kaewkes S, Mairiang E, et al. (2010) Opisthorchiasis and Opisthorchis-associated cholangiocarcinoma in Thailand and Laos. Acta Trop.